\documentstyle[12pt,twoside]{article}
\input{macros-course}

\begin{document}
\handout{quiz-info}{Quiz Information}{February 15, 1994}
\bigskip

\noindent
{\bf When:} Wednesday, March 2, 1994  at 7:00--9:00 {\sc p.m.}
\bigskip

\noindent
{\bf Where:} Rooms 4--149 and 4--153 (go to either).
%	April 6: Walker Memorial Gym, 50--340.
\bigskip

\noindent {\bf No lecture:}
There will be no lecture the day before the quiz, March~1.

\bigskip

\noindent {\bf Crib sheet:} The quiz is {\em closed book}.  
You may, however, bring {\em one\/} handwritten
$8\frac{1}{2}'' \times 11''$ sheet of paper
as a crib sheet.  Preparing a crib sheet can be a useful study aid, so
take time in selecting material for it. You may use both sides of the
paper and write as small as you like, but in any case, you are allowed
only {\em one\/} sheet.  

\bigskip

\noindent 
{\bf Material covered:} 
You are responsible for material
covered in lecture and problem sets 
before the quiz (through February~28).
You are also responsible for the material assigned as readings.

\bigskip

\noindent
{\bf Problem Set 4:}
Because there is no lecture March 1, 
both the oral and written parts of 
Problem set 4 will be due in tutorial Monday, February 28.
Graded problem sets will be available for you to pick up from
the course secretary, Cheryl Patton, in NE43--201 between 3:00 and
5:00~{\sc p.m.} on Tuesday, March~1.

\bigskip

\noindent 
{\bf Conflict:} 
If you are unable to attend the quiz
at the scheduled time, please talk to your TA in advance. 

\end{document}
